Author’s response to reviews

Title: Clinical Experience with Tigecycline in the Treatment of Hospital-Acquired Pneumonia Caused by Multidrug Resistant Acinetobacter baumannii

Authors:

Yangang Zhou (zhouyangang@csu.edu.cn)
Xumin Chen (1120340591@qq.com)
Ping Xu (xuping1109@csu.edu.cn)
Yan Zhu (zhuyan0918@csu.edu.cn)
Kuangguo Wang (aledith@126.com)
Daxiong Xiang (xiangdaxiong@163.com)
Feng Wang (roitech@126.com)
Hoan Banh (Hoan@ualberta.ca)

Version: 3 Date: 18 Feb 2019

Author’s response to reviews:

Dear editors,

Thank you very much for your feedback on “Clinical Experience with Tigecycline in the Treatment of Hospital-Acquired Pneumonia Caused by Multidrug Resistant Acinetobacter baumannii”, I am very appreciated about the suggestion you made. The manuscript was revised in accordance with BioMed Central editorial policies and formatting guidelines. Here are the responses to the reviewers.

Stefanie Krick (Reviewer 1):

1. 1. Line 228/229: please change "non-bronchial alveolar lavage" to "bronchoalveolar lavage"

Response: We have revised the sentence as you suggested and high lined in the revised manuscript, Line 229-230.

2. Line 269: please change to: tigecycline achieves a low plasma concentration at this dose
Response: We have revised the sentence as you suggested and high lined in the revised manuscript, Line 269.

Antonello Di Paolo, MD, PhD (Reviewer 2):

The Authors answered to criticisms and queries. The low number of patients still remains as the major limit of the study. A minor language revision is needed for some sentences

Response: Thank you for your suggestions. The language has been revised again.